Author's response to reviews

**Title:** Cold Saline Irrigation of the Renal Pelvis During Radiofrequency Ablation of a Central Renal Neoplasm

**Authors:**

Jose Pablo Morales (jpmorales@doctors.org.uk)
Tarun Sabharwal (Tarun.Sabharwal@gstt.nhs.uk)
Marios Georganas (gmarios@doctors.org.uk)
Renato Dourado (Renato.Dourado@gstt.nhs.uk)
Declan Cahill (Declan.Cahill@gstt.nhs.uk)
Andreas Adam (Andy.Adam@kcl.ac.uk)

**Version:** 11  **Date:** 1 February 2008

**Author's response to reviews:** see over
Dear Professor Michael Kidd,

Professor and Head of the Discipline of General Practice
The University of Sydney

[Address]

1 February 2008

Dear Editor,

Cold Saline Irrigation of the Renal Pelvis During Radiofrequency Ablation of a Central Renal Neoplasm

Authors: Jose P Morales, Tarun Sabharwal, Marios Georganas, Renato Dourado, Declan Cahill, Andreas Adam

Thank you very much for accepting our paper for publication in the Journal of Medical Case Reports. We have made all changes requested by The BioMed Central Editorial Production Team and the point-by-point description of those changes is at the end of this letter. Please find enclosed the edited version of our manuscript and figures.

We are very grateful that this opportunity was given to us for publishing in JMCR.

Yours sincerely,

The Authors
**PS:** Point-by-point changes made according to The BioMed Central Editorial Production Team

Title – The title was reformatted as required in both the manuscript and in the submission system. Capitals were only used in the first word and proper nouns. The full stop was removed from the end of the title.

Author list - The list of authors in the manuscript is now written exactly as they are in the submission system.

Qualifications – Those have been removed as requested.

Category/Abbreviated title – Those have been removed.

Abstract – The abstract has been structured as required. (Introduction, Case presentation, Conclusions). It was also updated on the submission page.

Case report -> Changed for Case presentation

Section headings – Changed as requested.

Please place the references and figure legends after the 'Editor's note'. Those have been placed after Editor’s note as requested.

Additional file – Removed as requested.

Figure order – Figure 3 now appears on the text. Page 3, paragraph 1, line 4.